Author’s response to reviews

Title: Influenza vaccination coverage against seasonal and pandemic influenza and their determinants in France: a cross-sectional survey

Authors:

Sophie Vaux (s.vaux@invs.sante.fr)
Dieter Van Cauteren (d.vancauteren@invs.sante.fr)
Jean-Paul Guthmann (jp.guthmann@invs.sante.fr)
Yann Le Strat (y.lestrat@invs.sante.fr)
Véronique Vaillant (v.vaillant@invs.sante.fr)
Henriette de Valk (h.devalk@invs.sante.fr)
Daniel Lévy-Bruhl (d.levybruhl@invs.sante.fr)

Version: 5 Date: 11 January 2011

Author’s response to reviews:

“Influenza vaccination coverage against seasonal and pandemic influenza and their determinants in France: a cross-sectional survey”

Dear Melissa Norton,

Please find enclosed the manuscript referenced above, which has been revised taking into account the editorial production team comments. We would like to submit this revised paper for publication as a research article in BMC Public Health.

Concerning major revision required: the first table referenced in the manuscript is labelled Table 1 (page 3), the second is labelled table 2 (page 8) and the third is labelled table 3 (page 9) and the tables are provided in this same order at the end of the manuscript file.

Concerning the figure, it is not possible to cropped more closely the figure under the Powerpoint format without changing the proportions of the figure.

That the reason why we provided the figure as an image with the TIF format.

We hope that these changes will satisfy the reviewers and look forward seeing the paper published in BMC Public Health.

Yours sincerely,

Please forward all correspondence regarding this manuscript to:
Dr Sophie Vaux
Département des maladies infectieuses
Institut de Veille Sanitaire
12 rue du Val d’Osne
94 415 Saint-Maurice Cedex
France
Phone : 33 (0)1 41 79 69 98
Fax : 33 (0)1 41 79 68 72
s.vaux@invs.sante.fr

Thank you for your consideration.
Sophie Vaux, PharmD